Contents lists available at ScienceDirect

Heliyon



journal homepage: www.cell.com/heliyon

Research article

5²CelPress

Commodity prices and economic growth: Empirical evidence from countries with different income groups

Ali Şen^a, Ahmet Gökçe Akpolat^b, İsmail Balkan^{b,*}

^a Faculty of Economics and Administrative Sciences, Department of Economics, Turkey
^b Kutahya Faculty of Applied Sciences, Department of Finance and Banking, Turkey

ARTICLE INFO

JEL classification: 013 Q33 C33 Keywords: Commodity prices Economic growth Panel system GMM method

ABSTRACT

The purpose of this paper is to examine the effect of global commodity prices such as beverage, energy, fertilizer, food, metal and mineral, precious metal and agricultural raw material on GDP per capita of countries with different income levels which are low, lower-middle, upper-middle, and high. The results of the study using panel system GMM method over the period 2007-2021 showed that for all income group countries, the impact of energy and fertilizer prices on GDP per capita is negative, while the impact of food and metal and mineral prices is positive on GDP per capita. The study also found that rising prices of agricultural raw materials reduces GDP per capita of all income group countries except lower-middle income countries. Moreover, according to the results of the study, rising beverage prices increased the GDP per capita only of highincome countries, while rising precious metal prices decreased the GDP per capita of lowermiddle and high-income countries. The study revealed that price changes in all commodity groups have an impact on the GDP per capita of high-income countries. It is demonstrated that price changes in all commodity groups have an impact in both directions on the GDP per capita of all income groups, depending on whether they are net producers or net consumers. The results of the study showed that, contrary to the literature, the countries most affected by commodity prices are high-income countries. Based on the empirical findings, this study point to the need for international cooperation to minimize the adverse effects of commodity price changes.

1. Introduction

Commodity prices, which do not follow a straight line, often fluctuate more than the prices of manufactured goods or services [1]. In commodity markets notorious for their volatility, it's common for prices to fluctuate by more than thirty percent in a year or two [2]. However, the roller coaster trend in commodity prices, which occurred rarely in the past, has recently been a recurring feature of global commodity markets [3,4]. Since the mid-1990s, prices in commodity markets have followed a more volatile course because of the financialization of commodity markets, increases in demand from the global manufacturing industry, fluctuations in factor input costs, unfavorable weather conditions, and intensifying geopolitical tensions [5,6]. Especially due to the disruptions in supply chains following the COVID-19 pandemic and Russia's invasion of Ukraine since the beginning of 2022, the prices of many commodities climbed rapidly and reached their highest levels in history since the 1990s (Fig. 1).

Recently, changing commodity prices has made an unambiguous effect on the income levels of all countries, no matter how

* Corresponding author. *E-mail addresses:* ali.sen@dpu.edu.tr (A. Şen), ahmet.akpolat@dpu.edu.tr (A.G. Akpolat), ismail.balkan@dpu.edu.tr (İ. Balkan).

https://doi.org/10.1016/j.heliyon.2024.e34038

Received 8 November 2023; Received in revised form 24 June 2024; Accepted 2 July 2024

Available online 3 July 2024

^{2405-8440/© 2024} Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

different their development levels are. Theoretically, commodity prices affect a country's income through two channels, "income" and "cost" [8]. Commodity prices determine the value of the commodity and affects the demand for commodities used in agricultural and manufacturing industrial production. According to Deaton's [9] net benefit approach, the effect of the price of a commodity on income depends on whether the country is a net producer or net consumer of the commodity in question. Net producers and exporters benefit from higher commodity prices, while net consumers and importers suffer. The reverse is true when commodity prices decline [10–14].

The dependence of export revenues of many developing countries, especially low-income ones, on a few commodities has been of interest for both academics and policy makers. Until the 1980s, abundance of natural resources was considered to be an important advantage for developing countries attempting to achieve a rapid economic growth. Since the 1980s, a sizable literature, challenging the view that an abundance of natural resources is a blessing for these countries whose exports are dependent on a small number of commodities, has emerged [15–20]. These studies have suggested that countries with abundant natural resources tend to grow slower than countries with scarce resources, and that natural resource endowment could turn into a "curse". However, many studies have also claimed that the abundance of natural resources is a blessing and positively affects economic growth [21–23]. Besides, in recent years, many studies have focused on the effect of commodity price volatility on economic growth. The overwhelming majority of those studies found that commodity price volatility negatively affects economic growth [22,24–26]. A few studies have pointed out the existence of a positive relationship between commodity price volatility and economic growth [27,28]. Empirical results of some studies have also revealed that there is no relationship between commodity price volatility and growth [29,30].

However, these studies have focused mostly on low-income countries whose income depends on the export of a small number of commodities [22,27,31]. It is difficult to argue that commodity price movements are only a problem for the low-income countries. In a globalizing world, no country, whatever its income level, is immune from the effects of commodity price movements [13]. In the past, commodities were viewed as plentiful and cheap for the growth of high-income countries. Today, those countries cannot easily access them due to increasing drought, wars, bottlenecks in production, and more autarkic anti-globalization policies implemented in the post-pandemic period [4]. Although high-income countries still account for the majority of global commodity consumption, especially emerging market countries, which have a strong growth performance, have also started to take a larger share of the commodity market since the early 2000s [32,33]. Therefore, the burning question is to reveal the impact of commodity price changes on the income of countries in all income groups. This study attempts to fill this gap in the literature.

The primary objective of this study is to investigate the impacts of commodity prices like energy, fertilizer, food, raw materials, metals/minerals, and precious metals on GDP per capita of countries with different income levels. Compared to studies in the existing literature, the study covers changes in prices of a wide range of commodities. Moreover, the study focuses on the impact of change in prices not only on the economic growth of low-income countries whose export incomes are dependent on a few commodities, but also on the economic growth of high-income countries, which are the largest consumers of the global commodity market. In the current literature, this is the only study, at least to the authers' knowledge, that examines countries together by considering all income groups separately. To this end, based on the world bank's country classification, we choose 168 countries from 4 different income groups, namely: low, lower-middle, upper-middle, and high.

This study contributes to the existing literature in four ways. Firstly, this study investigates the impact of changing commodity prices on the GDP per capita of countries with different income levels. The extensive literature has focused more on low-income countries that are dependent on a few commodities and other income group countries that import commodities. To the best of our knowledge, this study is novel in contributing to the literature analyzing countries from all income groups, both commodity exporters and commodity importers together. Secondly, this study has a very comprehensive sample of 168 countries. Although there are studies examining many countries in the literature [34–36]. This study is unique in the literature in terms of country sample size. Thirdly, in



Fig. 1. Commodity price index (1990–2023), source [7].

this study, all commodity types as seven main groups were used to investigate the effect of each commodity price on GDP per capita. Although there are studies that analyze the effect of commodity price volatility on growth, these studies include a limited number of commodities [30,37–39]. Lastly, although there are many studies in the literature examining the effects of commodity price volatility on economic growth, this study attempts to examine the impact of commodity prices levels on GDP per capita.

The rest of the study is organized as follows: Section 2 reviews the relevant literature. Section 3 presents the empirical methodology and data. Section 4 presents a discussion about the empirical results. Section 5 provides the conclusion and policy implications.

2. Literature review

One of the burning problems of recent times in the literature is related to the effects of commodity prices. Commodity prices affect both the production of the commodity and related goods and the consumption of commodity [37]. The relationship between commodity prices and economic growth has a significant place in the literature. The results of these studies vary depending on the type of commodity, country or region.

Beverage consisting of coffee, cocoa and tea is produced mostly by low-income countries in South America, Africa and South Asia. Therefore, the ups and downs in the prices of these products have the potential to affect the macroeconomic aggregates of the countries in question, such as growth, employment, poverty, exports and government revenue. There are many studies in the literature focusing on the impact of beverage exports on economic growth [40-45]. Additionally, some studies have examined the impact of beverage taxes on beverage prices [46–49]. However, there are few studies in the literature examining the impact of beverage prices on economic growth. In this regard [50], investigated whether commodity prices affect the real exchange rate and manufacturing industry production in Colombia, which is dependent on two different products, oil and coffee. They applied a vector error correction model using annual data from 1972 to 2013. According to the results, there is a positive relationship between commodity prices and real exchange rate, and therefore commodity price increases harm Colombia's competitiveness. In contrast, Bussolo et al. [51], using household survey data for Uganda, found that the increase in coffee prices was accompanied by a significant reduction in poverty. Similarly, by using regression model with time series from 1961 to 2012, Ankrah et al. [52] inveatigated whether world cocoa prices have an impact on cocoa production in Ghana. The findings of their study showed that the world cocoa price has a positive relationship with the production of cocoa in Ghana. Again, the study by Ref. [53] also showed that higher coffee prices lead higher household income and lower rates of poverty. However, Ofori-Abebrese et al. [54] reached very different results from previous three studies. Using autoregressive distributed lag model, they studied relationship between cocoa prices, exchange rate and economic growth for Cote d'Ivoire and Ghana over the period 1980-2011. According to the estimated results, the increase in cocoa prices negatively affects economic growth in Ghana only in the long term, but no significant relationship was found between the two variables in the short term. On the other hand, the study did not reach significant results for Cote d'Ivoire, both in the short and long term.

Fertilizer, one of the most important agricultural inputs, has a crucial role in increasing agricultural productivity. Fertilizer prices alone correspond to 15–20 % of the total costs in agricultural production [55]. Some studies in the literature have focused on analyzing the effects of fertilizer policies, such as pricing and subsidies [56–59]. On the other hand, most studies have examined more on the effects of fertilizer use on agricultural production. In this respect, a study by Ref. [60] tested the relationship between agronomic inputs such as fertilizer and cereal yields per hectare. The results of the study confirmed that the use of fertilizer increases cereal yield. They furher found that the increase in agricultural production in the China. Similarly [62], also demonstrated that fertilizer use stimulates land productivity and increases agricultural income. On the other hand, the study of [63] employed a simulation model to examine the impact of fertilizer prices on different indicators such as income of households in Malawi. The results revealed the existence of a negative relationship between fertilizer prices and income and other indicators [64]. found that rising fertilizer prices reduced the yields of most crops for 10 Asian countries over the period 1966–2005. Again, the study by Ref. [65] built different fertilizer price scenarios based on an econometric relationship with oil and gas prices from 2005 to 2050. According to the findings of their studies, rising fertilizer prices reduce crop yields.

Energy, which comprises coal, crude oil and natural gas, is one of the most fundamental inputs used in the production of different goods and services. Non-renewable energy sources still have a substential share in the total. Thus, rising energy prices will result in higher costs at all production stages throughout the country's economy. However, rising energy prices are to the benefit of net producer and exporting countries and to the detriment of net consumer and importing countries. A study by Arshad et al. [66] investigated whether energy prices affect economic growth in Pakistan. They used quarterly data for the period 1991-2011 and applied the Generalized Method of Moments (GMM). The estimated results revealed the negative impact of energy prices on growth. Similarly, Ferdaus et al. [67] investigated how energy consumption and energy prices affected economic growth for the Next 11 countries over 1990–2013. According to the results of panel Autoregressive Distributed Lag approach, energy prices negatively affect economic growth both in the long-run and short-run. Contrary to Arshad et al. [66] and Ferdaus et al. [67], Shahbaz et al. [20] examined the link between oil prices and economic growth by applying the VECM Granger causality approach for the case of USA. The estimation results demonstrated that oil prices positively affect economic growth in the long run. Consistent with Shahbaz et al. [20], Prabheesh and Laila [68], by applying the NARDL approach, examined the impact of crude oil and palm oil prices on GDP in Indonesia over the 2000–2019 period. The findings of the study confirmed the existence of a significant and positive relationship between both oil and palm oil prices and the country's output. In the same vein, in case of Azerbaijan, Mukhtarov et al. [69], using the Johansen cointegration and vector error correction model approach, analyzed the influence of oil prices on economic growth and other macroeconomic variables from 2005 to 2019.

Precious metals consisting of gold, silver and platinum are considered "safe havens" to protect against negative economic

conditions, especially where fluctuations in asset prices occur [70,71]. Especially gold is in high demand due to its value storage role. Annual gold demand reached 4741 tons in 2022, almost the same level as in 2011, when extraordinary investment demand was experienced [72]. However, the increase in demand for precious metals will be met by imports unless there is domestic production. There have been studies in the literature examining the relationship between gold production and gold prices and other macroeconomic aggregates such as growth. In this regard, a study was presented by Bildirici and Sonustun [73] examined whether gold and oil prices have an impact on the growth of some oil exporting countries. By using MS-VAR method for the period from 1979 to 2018, they noted the importance of gold and oil price volatility on economic growth. Other study by Bildirici and Gokmenoglu [74] investigated whether precious metal production had an impact on the growth of Australia, Canada, Mexico, the Philippines, Peru, South Africa and the USA. By using Markov-Switching-Vector Error Correcting model, they found that in the long run precious metal production affects growth, but the effect of precious metal production on growth varies from country to country. Li [37] tested the price volatility of commodities such as livestock, agriculture, coal, crude oil, and precious metal on the growth of China from 1981 to 2019. According to results of, there does exist a positive relationship between growth and commodity prices. Guan et al. [75] studied how volatility in oil and gold prices affects the growth in the top 10 oil-producing and 10 gold producing countries. The study, by employing the panel autoregressive distributed lag model, used the 2000-2020 period data of oil and gold producing countries. They concluded that volatility in both oil and gold prices has adverse effects on growth in the long run. Raza et al. [76], who used monthly data for the period 2000-2015, examined the impact of gold and oil prices and their related volatilities on stock markets in China, India, Brazil, Russia, South Africa, Mexico, Malaysia, Thailand, Chile and Indonesia. According to the results of the nonlinear ARDL approach, gold prices affected the stock markets of BRICS positively and others negatively. Oil prices have negatively affected the stock markets of all countries. On the other hand, oil and gold price volatility had a negative impact on stock markets of all countries.

The prices of metals and minerals, including aluminum, copper, iron ore, lead, nickel, tin and zinc, are important not only for consumers who use them widely in various sectors, but also for the producing countries [77]. Therefore, rising metal and mineral prices lead to an income-increasing effect for net producer and exporter countries and an income-decreasing effect for net consumer and importer countries. Although there is a large literature examining the impact of metal and mineral imports and exports on economic growth [78-81], the number of studies examining the effects of prices of different metal and mineral types is more limited. In this regard, a study of [36] used a panel-VAR approach and constructed a commodity price index that includes the prices of oil and gas and the prices of nine minerals. According to estimated results, rising prices promote the level of national income per capita. Similarly [82], also examined the impact of metal-mining prices on poverty in Chile utilizing 1998–2013 household data. They found that rising metal prices increased employment while reducing poverty. In addition [83], claimed in their studies that mineral resources contribute to the economic and social development of countries. On the other hand [84], used a dynamic computable general equilibrium model and showed that fluctuation in world iron ore prices has detrimental effects on growth and employment in China. A study by Ref. [85] using a dynamic stochastic general equilibrium (DSGE) model for Chile invesitigated the effect of rising copper prices on growth and the results proved the positive relationship between variables [86]. employing the Bayesian Model Averaging (BMA) approach over the period 1995–2015 for Zambia examined relationship between the copper price dynamics and economic growth. The study showed a positive and significant relationship between copper prices and economic growth [87], analyzed the relationship between copper prices and economic growth in Chile from 1995 to 2018. The results of the causality, cointegration and impulse-response function revealed the existence of a strong and positive relationship between copper prices and economic growth. By imlementing a quantile regression approach [88], examined the impacts of changing aluminium price movements on manufacturing industry production during the COVID period in India. According to estimated results, aluminium prices have a statistically significant and positive effects on Indian manufaturing sector.

Agriculturel raw materials, which comprise timber and other raw materials such as cotton, natural rubber and tobacco, are the primary source of many sectors, from textiles to housing and transportation. Most studies have focused on the impact of agricultural raw material production, imports and exports on economic growth [89–94]. But the current article has examined the impact of prices of aggricultural raw materials GDP per capita. Rising agricultural raw materials prices are a gain for the producer and exporter and a cost for the consumer and importer. There are few studies in the literature discussing the effects of agricultural raw material prices on economic growth. By using panel data model, GARCH model and impulse response function [95], investigated the interactions between prices and productions of natural rubber. The findings of the study emphasized that there is a positive relationship between price increases and economic growth for developing countries that export agricultural raw materials. Similarly [96], found that the rise in cotton prices would result in an increase in cotton production and farmers' income. On the other hand [97], analyzed whether the volatility in prices of agricultural raw materials affects the growth of developed countries. By based on the generalized autoregressive conditional heteroskedasticity (GARCH) model from 1980 to 2020, they found that agricultural raw material prices harm the economic growth of developed countries that import these raw materials. A study by Ref. [98] employed the two-step least-squares method to analyze the impact of volatility and the level of raw material prices on GDP and industrial production for groups of three different developed countries. According to the results, rising raw material prices improved income dynamics, while high price volatility worsened them.

Food, which consists of three different sub-groups such as creals, oils and meals, and other foods, are primary needs of humans and other living beings. If food prices rise, food producers and exporters earn higher incomes. Otherwise, food consumers and importers will have to bear higher costs [3]. In this regard, Ivanic and Martin [99] and Ivanic et al. [100] argued that rising in staple food prices in recent years increases the real incomes of food producers in many poor countries but have a detrimental impact on the incomes of net food consumers. They found that the impact of rising food prices on poverty varies by country and by commodity, and that the poverty-increasing effect was much more frequent and greater than the poverty-reducing effect. Wodon and Zaman [101] also reached similar results with those of the previous two studies and emphasized that higher food prices generally increase poverty in Sub-Saharan

Africa. On the other hand, Parabheesh and Laila [68] obtained different results. Using linear, and non-linear autoregressive distributed lag (ARDL) model for Indonesia in 2000–2019, they found that rising oil and palm oil prices posively affects on GDP. In line with Parabheesh and Laila [68], a study of Headey [102], using simulation models, tested the relationship between real food prices and change in poverty. Estimatied results revealed the existence of a relationship between food price increase and poverty reduction. The study also reported that agricultural production responded positively to food price increases. In other study by Headey and Hirnoven [103], who applied panel regressions and used annual data for 33 middle-income countries from 2000 to 2019, found that the increase in real food prices reduces the number of poor people and also encourages agricultural production, except countries with larger urban or non-agricultural populations.

3. Data and methodology

This study analyzes the impact of commodity prices on economic growth for the four different country groups over the period 2007–2021 using panel system GMM method. We take GDP per capita as a proxy to represent economic growth. We use seven different commodity prices which are the prices of beverages, energy, fertilizers, food, metals and minerals, precious metals and raw materials. In addition, we use four control variables to control for the effects of these variables. The control variables are the ratio of gross fixed capital formation to GDP, the ratio of international trade to GDP that is the ratio of sum of imports and exports to GDP, the ratio government expenditures to GDP and the population growth rate of each country. We take the natural logarithms of GDP per capita and commodity prices to obtain elasticity coefficients while we take level values of the control variables. The underlying reason for using variables in this way is that per capita GDP and commodity prices are expressed quantitatively, and control variables are expressed proportionally. To control for the effects of the Covid-19 pandemic, we use a dummy variable that takes a value of 1 for the years 2020 and 2021, and 0 for the other years. As is known, the Covid-19 pandemic manifested itself in the form of production declines, disruption of the supply chain, supply constraints and subsequently rising inflation in the majority of world economies. If these effects of Covid-19 are not controlled for by using a dummy variable, they may be misinterpreted as the effect of commodity prices and other control variables, which are explanatory variables, on national income per capita, the dependent variable of our analysis. Therefore, by using a dummy variable for Covid-19, it is aimed to control these effects and see the net effects of other explanatory variables.

Table 1 shows the descriptive statistics for the explanatory variables. The reason for removing the letter L from the names of the variables is that the actual values of the variables are used, not their logarithmic values.

Panel A in Table 1 shows the descriptive statistics of commodity prices. Since these prices are global, they are valid for all countries for each year. Mean, median and maximum values of FERT is the highest as compared with the other commodity prices. Looking at skewness, all the values of commodity prices are positively skewed. In other words, mean values of the commodity prices are more than the corresponding mode values and the data are skewed left. Kurtosis values indicate that only RAW, FERT an BEV have a leptokurtic distribution since their kurtosis values are 4.724, 3.667, 3.249 and they are greater than 3 while the other commodity prices are platykurtic since their values are lower than 3. This can be interpreted that while RAW, FERT and BEV have a relatively high probability of extreme events, whereas the opposite is true for the other commodity prices which have platykurtic distributions. In other words, RAW, FERT and BEV have many outliers while the other commodity prices have fewer outliers.

Looking at panel B showing the descriptive statistics of GDPC for each country group, the highest standard deviation is of highincome country group while the lowest one is of upper-middle income group. In other words, the most heterogenous group in terms of income per capita is high-income country group while the most homogeneous one is upper-middle income group. As for skewness, all the country groups have a positively skewed distribution This shows that the data all income groups have a skewness to

Table 1

Descriptive statistics.

Panel A: Descriptive Statistics of Commodity Prices							
	BEV	ENERGY	FERT	FOOD	METMIN	PREC	RAW
Mean	88.64131	95.42059	107.1733	101.1170	88.56996	104.2237	88.41558
Median	86.09782	94.36915	100.0000	93.12163	84.79044	100.0000	82.94613
Maximum	115.9746	135.2151	190.0851	124.1511	116.4445	140.1723	122.0384
Minimum	68.71736	52.68381	73.19746	83.86603	62.96807	59.49352	75.79315
Std. Dev.	11.33417	27.90660	31.52155	13.89191	16.41600	23.81169	11.82132
Skewness	0.583151	0.094141	1.081523	0.488755	0.161532	0.017099	1.474400
Kurtosis	3.248816	1.676597	3.666632	1.663571	1.870587	2.149437	4.724461
Panel B:Descriptive Statistics for GDPC of Different Country Groups							
	Low Income		Lower-Middle Income		Upper-Middle Income		High Income
Mean	702.65	14	2360.931		6897.185		40120.67
Median	640.6767		1998.738		6238.110		32169.50
Maximum	3188.750		9225.845		19849.72		234315.5
Minimum	170.70	69	321.8283		1531.222		6205.822
Std. Dev.	380.19	49	1398.693		2864.519		29271.33
Skewness	2.565324		1.554616		1.075931		2.375717
Kurtosis	12.920	32	6.787731		4.452530		11.40031

the left and the mean values are more than the corresponding mode values. Kurtosis values demonstrate that all the country groups have a leptokurtic distribution since their kurtosis values are higher than 3 and 12.920, 6.788, 4.452 and 11.400 for low-income, lower-middle income, upper-middle income and high income groups, respectively. This can be interpreted that the data of all country groups have a relatively high probability of extreme events. Moreover, all the country groups have many outliers.

3.1. Panel system GMM regression model

3.1.1. Dynamic panel data analysis and panel system GMM technique

Panel data analysis is key in identifying relationships that are difficult to detect with a particular cross-section or time series. In addition, panel data analysis emerges as a method with a wide range of uses, as it has more degrees of freedom, is based on more data, and eliminates multicollinearity problems to a large extent compared to time series models. However, the behavior patterns of economic actors can be affected by the behavior of the previous period(s) due to many reasons such as permanence or habit persistence. For this reason, dynamic modeling is needed in many panel data analysis. Many economic activities appear to be inherently dynamic. For this reason, dynamic modeling in panel data analysis provides researchers to comprehend the adjustment dynamics more clearly [104]. Balestra and Nerlove [105] suggested the first panel data model. Since then, panel data models have become very popular in the literature [106].

A general panel data model is given by:

$$y_{it} = \delta y_{i,t-1} + x'_{it}\beta + u_{it} \ i = 1, \dots, N; t = 1, \dots, T$$
(1)

In equation (1), y_{it} is dependent variable and y_{it-1} is the lagged dependent variable, respectively. x_{it} represents a vector of regressors while u_{it} symbolizes the disturbance term. The error term takes the form of a one-way error component model and can be represented as:

$$u_{it} = \mu_i + \nu_{it} \tag{2}$$

where the μ_i and ν_{it} have a form of normal distribution in equation (2). On the other hand, several issues arise due to the use of the lagged dependent variable as a regressor. Due to the fact that y_{it} is a function of μ_i , its lagged value is also a function of μ_i . For this reason, y_{it-1} becomes associated with the error term. Since the lagged value of the dependent variable is associated with the error term, it makes the OLS estimator biased and inconsistent, even if ν_{it} is not serially correlated. The fixed effects estimator is consistent only in the case that the time dimension is too large as compared with the number of cross-sections. A method is suggested by Anderson and Hsiao [107] that provides individual effects to be removed via the first difference transformation. The argument that instrumental variables can be used for the estimation of the difference equation is suggested by Anderson and Hsiao [107]. On the other hand, this method results in inefficient estimates, even if they are consistent, as they do not satisfy all the appropriate moment conditions [104]. The fact that not using all possible instrumental variables is a reason for this situation is stated by Arellano and Bond [108]. Therefore, they suggested the generalized method of moments (GMM). The employment of all acceptable previous values of variables as instruments is an effective technique is suggested by Arellano and Bond [108]. They suggested a similar way with Anderson and Hsiao [107] to remove the effects that are idiosyncratic to each cross-section. On the other hand, they showed the use of the difference equation is associated with ($Y_{i,t-1} - Y_{i,t-2}$). If the error term does not have serial correlation and the regressors are weak, it is supposed that the following moment conditions hold:

$$E[Y_{i,t} - s(\nu_{it} - \nu_{i,t-1})] = 0 \text{ for } s \ge 2; t = 3, \dots, T$$

$$E[X_{i,t} - s(\nu_{it} - \nu_{i,t-1})] = 0 \text{ for } s \ge 2; t = 3, \dots, T$$
(3)

where X's represent the independent variables ($v_{it} - v_{it-1}$) and denotes the new disturbance term in equation (3).

Since they are commonly used in dynamic panel data analysis, Soto [109] emphasized that both the difference GMM and the system GMM were developed within the context of labor and industrial studies which the number of individuals (N) is very large. Nevertheless, the large number of dynamic panel studies involve less not more than 100 cross-sections and probably less than half that. In addition, not all instrumental variables may be employed because of the small sample size, which may reduce the effectiveness of the various GMM estimators used in the research. Therefore, Soto [109] examined the efficiency of the several estimators involving applying Monte Carlo simulations to small samples. The fact that system GMM estimator has the lowest bias relative to the other estimators (i.e the difference GMM estimator) is the key finding of the study [109]. In addition, Blundell and Bond [110] put emphasize on that the conventionally used difference GMM estimator acquired by first-differencing transformation was determined to have large finite sample bias and poor precision characteristic in simulation studies in the case that the coefficient of the lagged dependent variable is large and the number of time series observations is small. Blundell and Bond [110] showed that the system GMM estimator that is proposed by the authors provides more effective results as compared with the difference GMM estimator under the condition that the additional moments conditions required for the system GMM are valid [111]. Therefore, the system GMM estimator is preferred to use in this study.

A forward orthogonal deviation transformation method in order to remove country-specific effects as compared to first-differencing transformation is developed by Arellano and Bover [108] due to the fact that difference transformation enlarge the gaps in the data. In the case of missing of one period of data, two differences are missing in transformed data. The forward orthogonal deviation

A. Şen et al.

transformation method makes changes to the data by subtracting each observation value from the mean of all future observation values, thus avoiding data loss no matter how many gaps there are. Owing to the fact that lagged observations are not included in the formula, they are considered as acceptable instrumental variables [112]. In addition, Hayakawa [113], using the Monte Carlo simulations method, showed that the GMM estimator of the model transformed with forward orthogonal deviation performs more efficiently than the first difference transformation. Hence, the forward orthogonal deviation transformation is selected to remove the effect that are idiosyncratic to each cross-section owing to its superiorities.

The econometric model adopted in this study is as follows:

$$LGDPC_{i,t} = \beta 0 + \beta_1 LGDPC_{i,t}(-1) + \beta_2 LBEV_t + \beta_3 LENERGY_t + \beta_4 LFERT_t + \beta_5 LFOOD_t + \beta_6 LMETMIN_t + \beta_7 LPREC_t + \beta_8 LRAW_t + \beta_9 GFCF_{i,t} + \beta_{10} TRADE_{i,t} + \beta_{11} GOV_{i,t} + \beta_{12} POP_{i,t}$$
(4)

where LGPDC and LGPDC (-1) natural logarithms of national income per capita and its lagged value respectively. LBEV, LENERGY, LFERT, LFOOD, LMETMIN, LPEC and LRAW are the independent variables of interest which are the prices of beverages, energy, fertilizers, food, metals and minerals, precious metals and raw materials, respectively. GFCF, TRADE, GOV and POP are the control variables which are the ratio gross fixed capital formation to GDP, the ratio of foreign trade volume (exports plus imports) to GDP, the ratio of government expenditures to GDP and population growth rate, respectively. Since the commodity prices subject to the research are global prices, they do not differ from country to country. On the other hand, the control variables are those that differ from country to country.

Table 2 presents sufficient information about the variables used in the research. However, it should also be noted that commodity prices are arranged as indices. While making this arrangement, the price of each commodity for 2010 was determined as 100. Prices before and after this date have been adjusted for current values in US dollars.

Since the panel GMM method is intrinsically suitable for short time intervals (T) and large cross-sections (N), unit root tests are generally not used as a preliminary test in panel GMM studies [114,115]. The main reason for this is that unit root tests can not give reliable results for short-term panel data. For this reason, we prefer not to use panel unit root tests as a preliminary. The following sub-section uses the data of the series in level rather than the differenced data.

3.1.2. Econometric results

Table 3 shows panel system GMM regression results. The first column shows the results of the low-income countries, while the other columns show the results of lower-middle, upper-middle and high-income countries.

Looking at the lagged dependent variable (LGDPC(-1)), it is positive and statistically significant for all country groups. It is seen that the autoregressive coefficient is 0.97 and 0.962 for the low-income and the lower-middle income country groups which are the highest ones. This shows that 1 % increase in the lagged income per capita leads to a 0.97 % and 0.962 % increase in the current income per capita for these country groups. On the other hand, the autoregressive coefficient (LGDPC(-1)) is 0.865 and 0.836 for the high-income and the upper-middle income country groups. In other words, a1% increase in the one-period lagged income per capita associates with a 0.865 and 0.836 increase current income per capita. These results show that the transition effect from the previous period to the current period per capita income is highest in low-income and lower-middle income countries, while it is lower in high-income and upper-middle income countries.

The coefficient of LBEV variable is statistically significant only for high-income country groups while it is not statistically significant for the other country groups. Looking at the coefficient, it is 0.10 for high-income country group. Accordingly, a 1 % increase in beverage prices is associated with a 0.10 % increase in per capita income in the high-income country group.

The coefficient of the LENERGY variable is negative and significant for all country groups. The coefficient of this variable is -0.181, -0.193, -0.249 and -0.190, respectively. That is, a 1 % increase in energy prices causes a decrease of 0.181 %, 0.193 %, 0.249 % and 0.190 %, respectively, in these country groups. These results clearly show that the increase in energy prices has a reducing effect on income in all country groups. In addition, the country group with the highest reducing effect is the middle-upper country group. In other country groups, the income-reducing effect is close to each other.

The coefficient of the LFERT variable is also negative for all country groups, similar to the coefficient of the LENERGY variable. The

Table 1	2
---------	---

Description of variables.

Symbol	Variable	Tranformation	Source
LGDPC	GDP per capita (Current US\$)	Natural Logarithmic	Worldbank-World Development Indicators
LBEV	Prices of Beverages	Natural Logarithmic	World Bank Commodity Price Data
LENERGY	Prices of Energy	Natural Logarithmic	World Bank Commodity Price Data
LFERT	Prices of Fertilizers	Natural Logarithmic	World Bank Commodity Price Data
LFOOD	Prices of Food	Natural Logarithmic	World Bank Commodity Price Data
LMETMIN	Prices of Metals and Minerals	Natural Logarithmic	World Bank Commodity Price Data
LPREC	Prices of Precious Metals	Natural Logarithmic	World Bank Commodity Price Data
LRAW	Prices of Raw Materials	Natural Logarithmic	World Bank Commodity Price Data
GFCF	Ratio of Gross Fixed Capital Formation to GDP	_	Worldbank-World Development Indicators
TRADE	Ratio of Trade to GDP	_	Worldbank-World Development Indicators
GOV	Ratio of Government Expenditures to GDP	-	Worldbank-World Development Indicators
POP	Population Growth Rate	_	Worldbank-World Development Indicators

Table-3

Panel system GMM regression results.

Explanatory Variables	Low-Income Countries	Lower-Middle Income Countries	Upper-Middle Income Countries	High-Income Countries
LGDPC(-1)	0.971** (0.470)	0.962*** (0.078)	0.836*** (0.129)	0.865*** (0.049)
LBEV	0.220 (0.185)	-0.017 (0.078)	0.078 (0.079)	0.10** (0.040)
LENERGY	-0.181** (0.086)	-0.193*** (0.065)	-0.249*** (0.071)	-0.190*** (0.026)
LFERT	-0.255)** (0.119)	-0.099** (0.038)	-0.216*** (0.032)	-0.253*** (0.028)
LFOOD	0.716** (0.285)	0.507*** (0.155)	0.854*** (0.122)	0.828*** (0.085)
LMETMIN	0.467*** (0.154)	0.511*** (0.089)	0.780*** (0.078)	0.632*** (0.042)
LPREC	0.020 (0.127)	-0.182** (0.087)	-0.139 (0.129)	-0.170*** (0.044)
LRAW	-0.399* (0.216)	-0.066 (0.139)	-0.556*** (0.208)	-0.517*** (0.081)
GFCF	0.043 (2.24)	0.0085 (0.0061)	-0.046 (0.177)	-0.239 (0.162)
TRADE	0.056 (0.039)	0.0052 (0.0035)	-0.041 (0.084)	-0.025 (0.030)
GOV	-0.127 (0.981)	-0.01** (0.005)	-1.092*** (0.199)	-1.646*** (0.386)
POP	-2.757 (6.22)	-2.178** (1.069)	-1.08** (0.541)	-0.931*** (0.238)
Covid-19 Dummy	-0.206*** (0.072)	-0.126*** (0.034)	-0.260*** (0.032)	-0.161*** (0.023)
Constant	-2.41 (3.65)	-1.69** (0.84)	-0.754 (1.52)	-0.116 (0.683)
AR(1)(p-value)	0.012	0.019	0.000	0.000
AR(2)(p-value)	0.740	0.132	0.989	0.254
Hansen J-test (p- value)	0.699	0.268	0.153	0.102
Number of Instruments	16	45	26	55
Number of Groups	19	51	43	55
Number of Observations	217	714	589	739

- Robust standard errors are reported in parentheses of estimated coefficients. The regression coefficients are estimated using the Arellano and Bover [116] and Blundell and Bond [110] system GMM approach.

- ***, *** and * represent statistically significance at 1 %, 5 % and 10 % significance level, respectively.

-Syntax used to obtain estimates of GMM is xtabond2 in Stata [117].

-The values reported for the Hansen J-test is the p-values for the null hypothesis of instrument validity. The values reported for AR(1) and AR(2) are the p-values for first-and second-order auto-correlated disturbances in the first differences equations.

coefficient of the LFERT variable is -0.255, -0.099, -0.216 and -0.253 for country groups, respectively. The coefficient of the LFERT variable is -0.255, -0.099, -0.216 and -0.253 for country groups, respectively. That is, a 1 % increase in fertilizer prices is associated with a -0.253 %, -0.099, -0.216 % and -0.253 % decrease in per capita income for these country groups, respectively. The income-reducing effect of fertilizer prices is evident in the low-income and high-income income groups, and this effect is very close to each other. On the other hand, while this reducing effect is slightly lower in the upper-middle income group, it is the least in the lower-middle income group.

As for LFOOD, the coefficient of this variable is 0.716, 0.507, 0.854 and 0.828, respectively, by country group. In other words, a 1 % increase in food prices causes an increase of 0.716 %, 0.507 %, 0.854 % and 0.828 % in these country groups, respectively. Incomeincreasing effect of food prices is highest in upper-middle income and high-income country groups. While this effect is slightly lower in the low-income country group, this effect is the lowest in the lower-middle income country group.

The LMETMIN variable has a positive coefficient for all country groups, similar to the LFOOD variable. The coefficient of LMETMIN is 0.467, 0.511, 0.780 and 0.632, respectively, by country group. This shows that a 1 % increase in metal and mineral prices causes an increase of is %0.467, %0.511,% 0.780 and %0.632 in these country groups, respectively. Similar to food prices, income-increasing effect of metal and mineral prices is highest in upper-middle income and high-income country groups while this effect is lower in low-income and lower-middle groups, respectively.

LPREC variable has statistically significant coefficients in the lower-middle and the high-income country groups while the other country groups do not have a statistically significant sign. The coefficient of lower-middle income and high-income group is -0.182 and -0.170, showing that a 1 % increase in the precious metals leads to a -0.182 % and -0.170 % decrease in the income per capita for these country groups, respectively.

As for LRAW, the coefficients of all the country groups are negative. The coefficients of the upper-middle-income and high-income country groups have a statistically significant sign at the 1 % significance level, and these coefficients are -0.556 and -0.517, respectively. On the other hand, the coefficient of low-income group is -0.399 and significant at 10 % significance level while the coefficient of lower-middle income group is statistically insignificant. According to these results, a 1 % increase in the raw material prices associates with a -0.556 %, -0.517 % and -0.399 % decrease in the income per capita for the upper-middle income, high-income and low-income country groups, respectively.

GFCF, GOV, TRADE and POP determined as control variables are among the most widely adopted control variables in the literature. The reason why these variables are among the most adopted variables is that they are considered to be among the factors that determine and affect the national income per capita. Even if the statistically significant effects of these variables could not be determined as a result of the study, it does not mean that these variables are incorrectly determined variables. The most important reason for including these variables in the study is to purify the effects of these control variables. GFCF and TRADE do not have a statistically significant sign for all country groups. As for GOV and POP, the coefficients of low-income country group are insignificant while they are negative and significant at 1 % significance level for the other country groups. As for POP, the significant coefficients are -2.178, -1.08 and -0.931 for the lower-middle, upper-middle and high-income country groups, respectively. The two interpretations

can be made in this case: First of all, the negative coefficients are compatible with the growth theory [118] showing that population growth rate negatively affects the economic growth. The coefficient of the POP variable is -2.178, -1.08 and -0.931 for the lower-middle, upper-middle and high-income country groups, respectively. According to these, when the population increases by 1 %, per capita income in these country groups decreases by 2.178 %, 1.08 % and 0.931 %, respectively. This result shows that the population growth rate has the highest income-reducing effect in the lower-middle income country group. This effect is less in the other country groups than in the low-income country group. GOV is a variable representing the ratio of public expenditures to national income, as described in Table 1. The coefficients of this variable are -0.01, -1.092 and -1.646 in the lower-middle income, upper-middle income reduces the per capita income by 0.01 %, 1.092 % and 1.646 %, respectively, in these country groups. In other words, it is determined for these three country groups that the ratio of public expenditures to national income has a decreasing effect on per capita income. In addition, it is seen that this reducing effect is higher in the high-income country group, lower in the upper-middle country group, and very low in the lower-income country group.

4. Discussion

The study results revealed the existence of a positive relationship between beverage prices and GDP per capita only for high-income countries. However, for other income group countries, there is no significant relationship between the two variables. These results are consistent with [54] who analyzed relationship between cocoa prices, exchange rate and economic growth for Cote d'Ivoire and Ghana. They found a relationship between cocoa prices and economic growth in the long term only for Ghana, but did not find a significant relationship between the two variables for Cote d'Ivoire in both the short and long term. Even though low-income countries are leading producers of coffee and cocoa, they have been unable to turn these products into drivers of growth enhancement and poverty reduction due to unfavorable internal and external conditions. However, coffee and cocoa imported from Africa and South America are re-exported by European countries with very high mark-up [118]. Although the top 5 coffee producing countries are Brazil, Vietnam, Colombia, Indonesia and Honduras, the share of raw common coffee bean exports of Germany, Switzerland, Italy, the USA, France, the Netherlands and Canada in 2002 is close to 50 % of the total. Similarly, while Ivory Coast and Ghana produce more than 60 % of global cocoa production, approximately 70 % of world cocoa bean exports in 2022 are carried out by just two countries, the Netherlands and Belgium [119]. On the other hand, the results of our study contradict the findings of Bussolo et al. [51] and Moss et al. [53], who observed that increasing coffee prices reduce poverty.

According to the results of the study, fertilizer prices negatively affect GDP per capita for all income group countries. The increase in fertilizer prices will reduce the use of the said input in production, which will diminish agricultural production and GDP. Both the coefficient and the significance level are higher for high-income and upper-middle-income countries with higher fertilizer adoption and application rates. These results align with the study of [63] who revealed the existence of a negative relationship between fertilizer prices and income and other indicators. Additionally, the results of both [64,65] are consistent with those of our study. In contrast, our results are not consistent with the study by Ref. [60], who found that the rising in fertilizer prices increases GDP per capita.

The study results have also indicated a negative association between energy prices and GDP per capita for all income group countries. The fact that the majority of countries are net energy consumers and importers matches the empirical results of the study. The results show consistency with [66,67], who examined whether energy prices affect economic growth and showed the negative impact of energy prices on growth. However, these results contradict the studies by Refs. [20,68], who revealed that oil price increases have a positive impact on economic growth.

The study also showed a negative relationship between precious metal prices and GDP per capita only for lower middle income and high income countries. On the other hand, no significant relationship was found between precious metal prices and GDP per capita for other countries. Although the booming gold demand has increased the income of a small number of gold-exporting countries, it has had a decreasing effect on the income of gold-importing countries, the majority of which have high incomes. The fact that there is a strong and negative relationship for especially high-income countries confirms that precious metals are also an important alternative investment tool for them. A study by Raza et al. [76] reached findings that were both consistent and inconsistent with the results of our study. They found that gold prices positively affect the stock markets of BRICS but negatively affect the stock markets of Mexico, Malaysia, Thailand, Chile and Indonesia.

This study confirmed the existence of a strong and positive relationship between metal/mineral prices and GDP per capita for all country income groups. These results indicate that for all country groups, the income gain effect of rising metal and mineral prices outweighs their income loss effect. The results of the study are consistent with those of studies by Refs. [36,82] who showed that rising prices promote the level of national income per capita and employment while reducing poverty. Similarly, except for the results of [84], the results of [85–88] are consistent with the results of our study.

The results also showed the significant adverse impact of agricultural raw material prices on GDP per capita for all income group countries except lower-middle income ones. Especially there is a strong association between agricultural raw prices and GDP per capita for upper-middle and high income countries, most of which are importers of agricultural raw materials. These results agree with the study of [98], who investigated the impact of volatility and the level of raw material prices on GDP, and found that rising raw material prices improved income dynamics, while high price volatility worsened them. Although the results of [97] coincide with the results of our study [95,96], produced results that were not compatible with our study.

Another result of the study is that there is a positive relationship between food prices and GDP per capita for all country income groups. Deaton [9] argued the effect of food prices depends on whether households are net consumers or net producers of food. In this regard, our results are in line with literature. Our results showed that the benefit of a rise in food prices to net producers outweighs the

harm to net consumers. In this regard, the results of the study are in line with studies by Headey [102], Headey and Hirnoven [103] and Parabheesh and Laila [68], who found that rising food prices stimulates the growth. On the other hand, our results are in contrast with those of Ivanic and Martin [99], Ivanic et al. [100] and Wodon and Zaman [101], who found that rising food prices have a poverty-increasing effect.

The fact that a significant and positive relationship between beverage prices and GDP per capita was found only for high-income countries does not support whether resource abundance is a "curse or blessing". The empirical results of negative impact of energy and fertilizer prices on GDP per capita for all country income groups also mean an increase in costs for both agricultural and industrial sectors. These findings do not confirm the existence of the "curse" or "blessing" hypothesis. The existence of a positive relationship between metal, mineral and food prices and per capita GDP for all country income groups does not provide information about whether the abundance of resources is "curse" or "blessing". Similarly, a negative relationship between precious metal prices and GDP per capita only for low-middle and high income group countries does not confirm the validity of either of the hypotheses in question. Moreover, the fact that agricultural raw material prices have a negative effect on GDP per capita does not indicate the existence of either hypothesis. These findings are a result of classifying countries according to their income levels, rather than the abundance of their resources.

5. Conclusion and policy implications

Global commodity prices have followed a much more up-and-down course over the last two decades compared to previous periods. Changing commodity prices have affected the macroeconomic fundamentals of all countries. Existing literature has focused on the impact of changing commodity prices on the growth of low-income countries whose income depends on exports of a small number of commodities. However, little is known about the impact of commodity prices on the incomes of countries with different income levels. Additionally, there is also a significant gap in the literature regarding the impact of changes in the prices of many commodity groups on the income of a large sample group consisting of many countries. This study aimed to examine the impact of changes in commodity prices on the GDP per capita of countries in different income groups. For this purpose, the study analyzed the impact of seven groups of commodity prices such as beverage, energy, fertilizer, food, precious metal, agricultural raw material, and metal and mineral on the GDP per capita of four groups of low, lower-middle, upper-middle and high income countries.

According to the results of the study, energy and fertilizer price increases led to a decrease in the GDP per capita of all income group countries. The fact that the overwhelming majority of countries are net consumers and importers of these inputs in both agricultural and industrial production is consistent with this result. On the other hand, the results of the study showed that the increase in food and metal and mineral prices led to an increase in GDP per capita for all groups. Countries that are net producers and exporters benefit from the increase in the prices of these commodities. The results also confirmed that the rise in prices of precious metals and agricultural raw materials negatively affects the GDP per capita of some country groups. Especially high-income countries have enhanced their import of precious metals, which they see as an investment tool and safe haven. Again, upper-middle and high-income countries have increased the import of agricultural raw materials to meet their textile, furniture and housing needs. In addition, the results of the study indicated that beverage prices positively impact GDP per capita of countries, from lower income groups to higher income groups. In particular, all commodity prices and GDP per capita of countries, from lower income groups to higher income groups. In particular, all commodity prices have significantly affected the GDP per capita of high-income countries. These results overlap with the aims of the study. On the other hand Classifying countries according to their income levels, rather than the abundance of their resources, does not confirm whether the "resource curse" or "resource blessing" hypotheses exist.

The results of our study have some implications for countries affected by changes in commodity prices. As markets become more integrated in a globalizing world, all countries, regardless of their income level, have been affected by the changes in commodity prices. Thus, first of all, there is a need for international policies that require cooperation and solidarity as well as national policies. The fact that especially the prices of energy, fertilizer and agricultural raw materials have an income-reducing effect for almost all country groups reveals the importance of international cooperation. Secondly, the policies aimed at preventing, minimizing and compensating for the effects of commodity price movements must also be coordinated and integrated with each other. Thirdly, the support of high-income countries and international organizations is vital, especially in compensating for the loss of low-income countries, many of which are located in Africa and South Asia.

Although this study obtained substantial empirical findings, it is held limitations. This study covers a limited period of price data on beverage, energy, fertilizer, food, metal/mineral, agricultural raw materials, and precious metal. A more extended period may be the subject of future research. Moreover, future research could also investigate the impacts of commodity prices on other economic, financial, environmental, socio-cultural and even psychological indicators. In addition, the lack of sufficient studies in the literature on some independent variables such as beverage and agricultural raw materials offers opportunities for future research.

Data availability statement

Available on request.

CRediT authorship contribution statement

Ali Şen: Writing – original draft, Supervision, Resources, Methodology, Conceptualization. Ahmet Gökçe Akpolat: Writing –

original draft, Methodology, Formal analysis, Data curation, Conceptualization. **İsmail Balkan:** Writing – review & editing, Resources, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

References

- [1] J.A. Frankel, The Natural Resource Curse: a Survey (No. W15836), National Bureau of Economic Research, 2010.
- [2] G.A. Davis, J.E. Tilton, The resource curse, in: Natural Resources Forum, vol. 29, Blackwell Publishing, Ltd, Oxford, UK, 2005, August, pp. 233–242, 3.
- [3] H. Bredenkamp, J. Bersch, Commodity price volatility: impact and policy challenges for low-income countries. Commodity Price Volatility and Inclusive Growth in Low-Income Countries, International Monetary Fund, Washington, DC, 2012, pp. 55–67.
- [4] M. Mohammed, The Economics and Finance of Commodity Price Shocks, Routledge, 2022.
- [5] International Monetary Fund, World Economic Outlook; Rocky Recovery, 2023. April.
- [6] World Bank, Commodity Markets Outlook; Lower Prices, Little Relief, World Bank, April. Washington, DC, 2023.
- [7] https://www.worldbank.org/en/research/commodity-markets.
- [8] D. Ferraro, P.F. Peretto, Commodity prices and growth, Econ. J. 128 (616) (2018) 3242–3265.
- [9] A. Deaton, Rice prices and income distribution in Thailand: a non-parametric analysis, Econ. J. 99 (395) (1989) 1–37.
- [10] M. Asif, K.B. Khan, M.K. Anser, A.A. Nassani, M.M.Q. Abro, K. Zaman, Dynamic interaction between financial development and natural resources: Evaluating the 'Resource curse' hypothesis, Resour. Pol. 65 (2020) 101566.
- [11] C.R. Azzoni, J.J. Guilhoto, F.G. Silveira, T. Menezes, E.A. Haddad, M.M. Hasegawa, Commodity price changes and their impacts on poverty in Developing Countries: the Brazilian case, Stud. Reg. Sci. 39 (1) (2009) 131–147.
- [12] A. Deaton, Commodity prices and growth in Africa, J. Econ. Perspect. 13 (3) (1999) 23-40.
- [13] C. Estrades, M.I. Terra, Commodity prices, trade, and poverty in Uruguay, Food Policy 37 (1) (2012) 58–66.
- [14] D. Lederman, G. Porto, The price is not always right: on the impacts of commodity prices on households (and countries), The World Bank Research Observer 31 (1) (2016) 168–197.
- [15] A.H. Gelb, Oil Windfalls: Blessing or Curse? Oxford university press, New York, NY, 1988.
- [16] Richard M. Auty, Sustaining Development in Mineral Economies: the Resource Curse Thesis, Routledge, London and New York, 1993.
- [17] T. Gylfason, Natural resources, education, and economic development, Eur. Econ. Rev. 45 (4–6) (2001) 847–859.
- [18] S. Rahim, M. Murshed, S. Umarbeyli, D. Kirikkaleli, M. Ahmad, M. Tufail, S. Wahab, Do natural resources abundance and human capital development promote economic growth? A study on the resource curse hypothesis in Next Eleven countries, Resources, Environment and Sustainability 4 (2021) 100018.
- [19] J.D. Sachs, A.M. Warner, Natural resource abundance and economic growth", in: NBER Working Papers 5398, National Bureau of Economic Research, 1995.
- [20] M. Shahbaz, K. Ahmed, A.K. Tiwari, Z. Jiao, Resource curse hypothesis and role of oil prices in USA, Resour. Pol. 64 (2019) 101514.
 [20] F.M. Anner, P. Cheler, P.L. P.G. Level, The level are regionary characteristic for an analysis of the level of
- [21] F.M. Aragon, P. Chuhan-Pole, B.C. Land, The local economic impacts of resource abundance: what have we learned? World Bank Policy Research Working Paper 7263 (2015).
- [22] A. Hayat, M. Tahir, Natural resources volatility and economic growth: evidence from the resource-rich region, J. Risk Financ. Manag. 14 (2) (2021) 84.
- [23] K. Nawaz, A. Lahiani, D. Roubaud, Natural resources as blessings and finance-growth nexus: a bootstrap ARDL approach in an emerging economy, Resour. Pol. 60 (2019) 277–287.
- [24] Y. Cao, S. Xiang, Natural resources volatility and causal associations for BRICS countries: evidence from Covid-19 data, Resour. Pol. 80 (2023) 103165.
- [25] M. Deng, China economic performance and natural resources commodity prices volatility: evidence from China in COVID-19, Resour. Pol. 75 (2022) 102525.[26] R. Van Eyden, M. Difeto, R. Gupta, M.E. Wohar, Oil price volatility and economic growth: evidence from advanced economies using more than a century's
- data, Appl. Energy 233 (2019) 612–621.
- [27] A. Benramdane, Oil price volatility and economic growth in Algeria, Energy Sources B Energy Econ. Plann. 12 (4) (2017) 338-343.
- [28] J. Wen, N. Mughal, M. Kashif, V. Jain, C.S.R. Meza, P.T. Cong, Volatility in natural resources prices and economic performance: evidence from BRICS economies, Resour. Pol. 75 (2022) 102472.
- [29] R. Rosnawintang, T. Tajuddin, P. Adam, Y.P. Pasrun, L.O. Saidi, Effects of crude oil prices volatility, the internet and inflation on economic growth in ASEAN-5 countries: a panel autoregressive distributed lag approach, Int. J. Energy Econ. Pol. 11 (1) (2021) 15–21.
- [30] L. Sun, Y. Wang, Global economic performance and natural resources commodity prices volatility: evidence from pre and post COVID-19 era, Resour. Pol. 74 (2021) 102393.
- [31] O. Joya, Growth and volatility in resource-rich countries: does diversification help? Struct. Change Econ. Dynam. 35 (2015) 38-55.
- [32] World Bank, Global Economic Prospects: the Turning Of the Tide? June, World Bank, Washington, DC, 2018.
- [33] J. Baffes, W.C. Koh, P. Nagle, The Evolution of Commodity Markets over the Past Century, Commodity Markets: Evolution, Challenges, and Policies, 2022.
- [34] C. Arndt, X. Diao, P. Dorosh, K. Pauw, J. Thurlow, The Ukraine war and rising commodity prices: implications for developing countries, Global Food Secur. 36 (2023) 100680.
- [35] S. Mohtadi, D. Castells-Quintana, The distributional dimension of the resource curse: commodity price shocks and income inequality, Struct. Change Econ. Dynam. 59 (2021) 63–78.
- [36] T. McGregor, Commodity price shocks, growth and structural transformation in low-income countries, Q. Rev. Econ. Finance 65 (2017) 285-303.
- [37] L. Li, Commodity prices volatility and economic growth: empirical evidence from natural resources industries of China, Resour. Pol. 80 (2023) 103152.
- [38] Q. Ma, M. Zhang, S. Ali, D. Kirikkaleli, Z. Khan, Natural resources commodity prices volatility and economic performance: evidence from China pre and post COVID-19, Resour. Pol. 74 (2021) 102338.
- [39] U.A. Sheikh, M. Asad, Z. Ahmed, U. Mukhtar, Asymmetrical relationship between oil prices, gold prices, exchange rate, and stock prices during global financial crisis 2008: evidence from Pakistan, Cogent Economics & Finance 8 (1) (2020) 1757802.
- [40] N. Gizaw, J. Abafita, T.M. Merra, Impact of coffee exports on economic growth in Ethiopia; an empirical investigation, Cogent Economics & Finance 10 (1) (2022) 2041260.
- [41] G.A. Noula, L.S. Gustave, D.G. Munchunga, Impact of agricultural export on economic growth in Cameroon: case of banana, coffee and cocoa, International Journal of Business and Management Review 1 (2013) 44–71.
- [42] D. Ouma, T. Kimani, E. Manyasa, Agricultural trade and economic growth in east African community, African Journal of Economic Review 4 (2016) 203–221.
- [43] Z.H. Shikur, Agriculture and commodity export and its productivity and production in Sub-Saharan Africa: evidence from time series data of four countries, Reg. Sect. Econ. Stud. 22 (1) (2022) 61–76.
- [44] A. Siaw, Y. Jiang, R. Pickson, R. Dunya, Agricultural exports and economic growth: a disaggregated analysis for Ghana, Theor. Econ. Lett. 8 (2018) 2251–2270, https://doi.org/10.4236/tel.2018.811147.
- [45] N. Verter, Cocoa export performance in the world's largest producer, Bulgarian Journal of Agricultural Science 22 (5) (2016).
- [46] J.C. Jones-Smith, L.P. Walkinshaw, V.M. Oddo, M. Knox, M.L. Neuhouser, P.M. Hurvitz, N. Chan, Impact of a sweetened beverage tax on beverage prices in Seattle, WA, Econ. Hum. Biol. 39 (2020) 100917.

- [47] J. Leider, L.M. Powell, Longer-term impacts of the Oakland, California, sugar-sweetened beverage tax on prices and volume sold at two-years post-tax, Soc. Sci. Med. 292 (2022) 114537.
- [48] J. Petimar, L.A. Gibson, J. Yan, S.N. Bleich, N. Mitra, M.L. Trego, C.A. Roberto, Sustained impact of the Philadelphia beverage tax on beverage prices and sales over 2 years, Am. J. Prev. Med. 62 (6) (2022) 921–929.
- [49] L.M. Powell, J. Leider, The impact of Seattle's sweetened beverage tax on beverage prices and volume sold, Econ. Hum. Biol. 37 (2020) 100856.
- [50] P. Poncela, E. Senra, L.P. Sierra, Long-term links between raw materials prices, real exchange rate and relative de-industrialization in a commodity-dependent economy: empirical evidence of "Dutch disease" in Colombia, Empir. Econ. 52 (2017) 777–798.
- [51] M. Bussolo, O. Godart, J. Lay, R. Thiele, The impact of coffee price changes on rural households in Uganda, Agric. Econ. 37 (2-3) (2007) 293-303.
- [52] S. Ankrah, K.A. Nyantakyi, E. Dadey, Modeling the causal effect of world cocoa price on production of cocoa in Ghana, Universal Journal of Agricultural Research 2 (7) (2014) 264–271.
- [53] C.B. Moss, A. Lyambabaje, J.F. Oehmke, An economic evaluation of SPREAD on Rwanda's rural population, Appl. Econ. 49 (36) (2017) 3634-3644.
- [54] J. Ofori-Abebrese, R.B. Pickson, G. Ofori-Abebrese, Commodity prices, exchange rate and economic growth in West Africa: case study of Cote dIvoire and Ghana, J. Dev. Agric. Econ. 9 (9) (2017) 269–277.
- [55] D. Headey, S. Fan, Anatomy of a crisis: the causes and consequences of surging food prices, Agric. Econ. 39 (2008) 375–391.
- [56] A. Cameron, C. Derlagen, K. Pauw, Options for reducing fertilizer prices for smallholder farmers in Tanzania. Options for Reducing Fertilizer Prices for Smallholder Farmers in Tanzania, 2017.
- [57] B. Manos, M.A.A. Begum, M. Kamruzzaman, I. Nakou, J. Papathanasiou, Fertilizer price policy, the environment and farms behavior, J. Pol. Model. 29 (1) (2007) 87–97.
- [58] H. Takeshima, L.S.O. Liverpool-Tasie, Fertilizer subsidies, political influence and local food prices in sub-Saharan Africa: evidence from Nigeria, Food Policy 54 (2015) 11–24.
- [59] Z.Y. Wu, X.L. Feng, Y.M. Zhang, S.G. Fan, Repositioning fertilizer manufacturing subsidies for improving food security and reducing greenhouse gas emissions in China, J. Integr. Agric. (2023).
- [60] J.W. McArthur, G.C. McCord, Fertilizing growth: agricultural inputs and their effects in economic development, J. Dev. Econ. 127 (2017) 133-152.
- [61] W. Huang, L. Jiang, Efficiency performance of fertilizer use in arable agricultural production in China, China Agric. Econ. Rev. 11 (1) (2019) 52-69.
- [62] E. Martey, J.K. Kuwornu, J. Adjebeng-Danquah, Estimating the effect of mineral fertilizer use on Land productivity and income: evidence from Ghana, Land Use Pol. 85 (2019) 463–475.
- [63] A.M. Komarek, S. Drogue, R. Chenoune, J. Hawkins, S. Msangi, H. Belhouchette, G. Flichman, Agricultural household effects of fertilizer price changes for smallholder farmers in central Malawi, Agric. Syst. 154 (2017) 168–178.
- [64] K.S. Imai, R. Gaiha, G. Thapa, Supply response to changes in agricultural commodity prices in Asian countries, J. Asian Econ. 22 (1) (2011) 61–75.
- [65] T. Brunelle, P. Dumas, F. Souty, B. Dorin, F. Nadaud, Evaluating the impact of rising fertilizer prices on crop yields, Agric. Econ. 46 (5) (2015) 653–666. [66] A. Arshad, M. Zakaria, X. Junyang, Energy prices and economic growth in Pakistan: a macro-econometric analysis, Renew. Sustain. Energy Rev. 55 (2016)
- 25-33.
- [67] J. Ferdaus, B.K. Appiah, S.C. Majumder, A.A.A. Martial, A panel dynamic analysis on energy consumption, energy prices and economic growth in next 11 countries, Int. J. Energy Econ. Pol. 10 (6) (2020) 87–99.
- [68] K.P. Prabheesh, N. Laila, Asymmetric effect of crude oil and palm oil prices on economic growth: evidence from Indonesia, Buletin Ekonomi Moneter Dan Perbankan 23 (2) (2020) 253–268.
- [69] S. Mukhtarov, S. Aliyev, J. Zeynalov, The effects of oil prices on macroeconomic variables: evidence from Azerbaijan, Int. J. Energy Econ. Pol. 10 (1) (2020) 72–80.
- [70] D.G. Baur, T.K. McDermott, Is gold a safe haven? International evidence, J. Bank. Finance 34 (8) (2010) 1886–1898.
- [71] F. Capie, T.C. Mills, G. Wood, Gold as a hedge against the dollar, J. Int. Financ. Mark. Inst. Money 15 (4) (2005) 343-352.
- [72] https://www.gold.org/goldhub/research/gold-demand-trends/gold-demand-trends-full-year-2022.
- [73] M.E. Bildirici, F.O. Sonustun, The effects of oil and gold prices on oil-exporting countries, Energy Strategy Rev. 22 (2018) 290–302.
- [74] M.E. Bildirici, S.M. Gokmenoglu, Does resource curse really exist in precious metal producer countries? Nat. Resour. Res. 28 (4) (2019) 1403–1428.
- [75] L. Guan, W.W. Zhang, F. Ahmad, B. Naqvi, The volatility of natural resource prices and its impact on the economic growth for natural resource-dependent economies: a comparison of oil and gold dependent economies, Resour. Pol. 72 (2021) 102125.
- [76] N. Raza, S.J.H. Shahzad, A.K. Tiwari, M. Shahbaz, Asymmetric impact of gold, oil prices and their volatilities on stock prices of emerging markets, Resour. Pol. 49 (2016) 290–301.
- [77] A. Rossen, What are metal prices like? Co-movement, price cycles and long-run trends, Resour. Pol. 45 (2015) 255-276.
- [78] N. Hlavová, The impact of mineral resources on economic growth, Int. J. Arts Commer. 4 (6) (2015) 100–110.
- [79] T. Isaiah Zayone, S.R. Henneberry, R. Radmehr, Effects of agricultural, manufacturing, and mineral exports on Angola's economic growth, Energies 13 (6) (2020) 1494.
- [80] N. Mahonye, L. Mandishara, Mechanism between mining sector and economic growth in Zimbabwe, is it a resource curse? Environ. Econ. (6, Iss. 3) (2015) 81–92.
- [81] A.K. Sahoo, D. Sahoo, N.C. Sahu, Mining export, industrial production and economic growth: a cointegration and causality analysis for India, Resour. Pol. 42 (2014) 27–34.
- [82] R. Alvarez, Á. García-Marín, S. Ilabaca, Commodity price shocks and poverty reduction in Chile, Resour. Pol. 70 (2021) 101177.
- [83] M. Ericsson, O. Löf, Mining's contribution to national economies between 1996 and 2016, Mineral Economics 32 (2) (2019) 223–250.
- [84] W. La-fang, L. Ming-yong, Z. Bao-jun, The economic impacts of world iron ore price shock using MCHUGE, a dynamic CGE model of China, in: 2007 International Conference on Management Science and Engineering, IEEE, 2007, August, pp. 1298–1303.
- [85] J.P. Medina, C. Soto, Copper Price, Fiscal Policy and Business Cycle in Chile, vol. 458, Banco Central de Chile, Santiago, 2007.
- [86] S. Chikalipah, Does a meaningful relationship exist between copper prices and economic growth in Zambia? African J. Econ. Manag. Stud. 10 (1) (2019) 72–84.
- [87] M. Marañon, M. Kumral, Empirical analysis of Chile's copper boom and the Dutch Disease through causality and cointegration tests, Resour. Pol. 70 (2021) 101895.
- [88] K. Mehta, R. Sharma, V. Vyas, A quantile regression approach to study the impact of aluminium prices on manufacturing sector of India during COVID era, Mater. Today: Proc. 65 (2022) 3506–3511.
- [89] S. Bakari, Do agricultural raw materials imports cause agricultural growth? Empirical analysis from north Africa, Bulletin of Economic Theory and Analysis 4 (2) (2018) 65–77.
- [90] D. Chabi Simin Najib, C. Fei, A. Dilanchiev, S. Romaric, Modeling the impact of cotton production on economic development in Benin: a technological innovation perspective, Front. Environ. Sci. 10 (2022) 926350.
- [91] J. Li, P. Cheng, X. Qian, Do mineral rents endowment and agricultural raw material imports determine natural resource management in the United States? Resour. Pol. 85 (2023) 103918.
- [92] K. Mahmood, S. Munir, Agricultural exports and economic growth in Pakistan: an econometric reassessment, Qual. Quantity 52 (4) (2018) 1561–1574.
- [93] P. Teye-Gaga, C.K. Obeng, B. Afful, M.T. Boadu, Economic integration and survival of Ghana's manufactured and agriculture raw material exports, Heliyon 9 (1) (2023) e12723.
- [94] M. Zaheer Khan, A. Salam, A. Batool, Nexus between agriculture raw material exports, trade openness and economic growth of Pakistan, J. Agric. Res. 53 (4) (2015).
- [95] R. Liu, L. Yang, C. He, Analysis on the dynamics between price and production of natural rubber in the world, Journal of Southern Agriculture 50 (10) (2019) 2359–2367.

- [96] J.M. Alston, D.A. Sumner, H. Brunke, Impacts of Reductions in US Cotton Subsidies on West African Cotton Producers, Oxfam America, Boston, MA, 2007.[97] E. Krasnovskiy, E. Nemtchinova, O. Mityakova, S. Garnik, S. Gasanbekov, A. Ten, D. Mardas, Estimating the volatility of commodity prices indices for
- agricultural raw materials using the GARCH model, Indian Journal of Economics and Development 18 (3) (2022) 742-747.
- [98] U. Nikonenko, S. Hanushchyn, G. Boikivska, Y. Andriichuk, V. Kokhan, Influence of world commodity prices on the dynamics of income of exporting countries of natural resources under globalization, Bus. Theor. Pract. 21 (1) (2020) 440–451.
- [99] M. Ivanic, W. Martin, Implications of higher global food prices for poverty in low-income countries 1, Agric. Econ. 39 (2008) 405–416.
- [100] M. Ivanic, W. Martin, H. Zaman, Estimating the short-run poverty impacts of the 2010–11 surge in food prices, World Dev. 40 (11) (2012) 2302–2317.
- [101] Q. Wodon, H. Zaman, Higher food prices in Sub-Saharan Africa: poverty impact and policy responses, The World Bank Research Observer 25 (1) (2010) 157–176.
- [102] D.D. Headey, Food prices and poverty, World Bank Econ. Rev. 32 (3) (2018) 676-691.
- [103] D. Headey, K. Hirvonen, Higher food prices can reduce poverty and stimulate growth in food production, Nature Food (2023) 1-8.
- [104] B.H. Baltagi, Econometric Analysis of Panel Data, vol. 4, Wiley, Chichester, 2008.
- [105] P. Balestra, M. Nerlove, Pooling cross section and time series data in the estimation of a dynamic model: the demand for natural gas, Econometrica: J. Econom. Soc. (1966) 585–612.
- [106] D.L. Millimet, I.K. McDonough, Dynamic panel data models with irregular spacing: with an application to early childhood development, J. Appl. Econom. 32 (4) (2017) 725–743.
- [107] T.W. Anderson, C. Hsiao, Estimation of dynamic models with error components, J. Am. Stat. Assoc. 76 (375) (1981) 598-606.
- [108] M. Arellano, S. Bond, Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations, Rev. Econ. Stud. 58 (2) (1991) 277–297.
- [109] M. Soto, System GMM estimation with small sample, in: Universitat Autonoma de Barcelona Working Paper, 2009, pp. 1–26. Barcelona, Spain.
- [110] R. Blundell, S. Bond, Initial conditions and moment restrictions in dynamic panel data models, J. Econom. 87 (1) (1998) 115–143.
- [111] A. Castelló-Climent, On the distribution of education and democracy, J. Dev. Econ. 87 (2) (2008) 179-190.
- [112] D. Roodman, A note on the theme of too many instruments, Oxf. Bull. Econ. Stat. 71 (1) (2009) 135–158.
- [113] K. Hayakawa, First difference or forward orthogonal deviation-which transformation should be used in dynamic panel data models?: a simulation study, Econ. Bull. 29 (3) (2009) 2008–2017.
- [114] D.A. Dickey, W.A. Fuller, Likelihood ratio statistics for autoregressive time series with a unit root, Econometrica: J. Econom. Soc. (1981) 1057–1072.
- [115] W.H. Greene, Econometric Analysis, third ed., Prentice-Hall, Upper Saddle River, New Jersey, 1997.
- [116] M. Arellano, O. Bover, Another look at the instrumental variable estimation of error-components models, J. Econom. 68 (1) (1995) 29-51.
- [117] R.M. Solow, A contribution to the theory of economic growth, Q. J. Econ. 70 (1) (1956) 65–94.
- [118] A. Elsby, Creaming off commodity profits: europe's re-export boom and Africa's earnings crisis in the coffee and cocoa sectors, Rev. Afr. Polit. Econ. 47 (166) (2020) 638–650.
- [119] https://www.tridge.com/intelligences.